



REVIEW HISTORY

Paper: **“Crossing Over: A Narrative Inquiry of One Man’s Journey Across the Mexico/U.S. Border”**

Corresponding Author: Beth Garcia

Email: egarcia@wtamu.edu

Doi: 10.19044/llc.v6no3a2

Peer review:

Reviewer 1: Blinded

Reviewer 2: Blinded

Published: 30.09.2019

